Author’s response to reviews

Title: Treatment Outcome and Its Predictors among Patients of Acute Heart Failure at a Tertiary Care Hospital in Ethiopia A Prospective Observational Study

Authors:

Mulubirhan Tirfe (mulubirhantirfe@gmail.com)
Teshome Nedi (teshome.nedi@aau.edu.et)
Desalew Mekonnen (desalewm@yahoo.com)
Alemseged Berha (alembeyene98@gmail.com)

Version: 3 Date: 06 Nov 2019

Author’s response to reviews:

Editors’ Comments and Authors Response

Manuscript Number: BCAR-D-18-00737
Manuscript Title: - Treatment Outcome and Its Predictors among Patients of Acute Heart Failure at a Tertiary Care Hospital in Ethiopia: A Prospective Observational Study
Authors: - Mulubirhan Tirfe; Teshome Nedi; Desalew Mekonnen; Alemseged Beyene Berha
We would like to thank the reviewer for a careful and thorough reading of this manuscript and for the thoughtful comments and constructive suggestions, which help to improve the quality of this manuscript. We have revised the manuscript as advised and changes made are shown in track changes. Point-by-point response for the reviewer’s comment is given below. We hope the revision undertaken has improved the manuscript to a level of your satisfaction and we request your editorial hand.
Response to Editor Comments

Comment 1:– “Studies must demonstrate medical/educational value of their findings to at least a regional/international audience instead of focusing conclusions to one hospital. Please address this in a revision by explicitly modifying aims, title, and discussion/conclusions."

Response 1:– As per the suggestion, we have made a correction.